

© The Author(s), 2022. Published by Cambridge University Press on behalf of The British Institute of International and Comparative Law. This is an Open Access article, distributed under the terms of the Creative Commons Attribution licence (http://creativecommons.org/licenses/by/4.0), which permits unrestricted re- use, distribution and reproduction, provided the original article is properly cited.

INDIRECT EXPROPRIATION AND THE PROTECTION OF PUBLIC INTERESTS

CHRISTIAN RIFFEL*

Abstract This article analyses what is widely known as the police powers exemption found in modern international investment agreements, with a focus on mega-regional trade and investment agreements. It explores its legal nature and requirements, the burden of proof and issues of compensation. In an attempt to curb (indirect) expropriation claims, the exemption carves out non-discriminatory regulatory measures from the scope of indirect expropriation and in such situations no compensation needs to be paid. It follows that, as a rule, foreign investors are not protected against the adverse economic effects of regulatory measures. The key question addressed is whether host States can increase the level of protection given to public welfare objectives through the use of this exemption without having to compensate investors for the measures taken. The article argues that, under the proportionality test embodied in the exemption, States can provide the level of protection that they desire without incurring a risk of liability as regards expropriation claims.

Keywords: police powers, indirect expropriation, rare circumstances, burden of proof, compensation.

I. INTRODUCTION

Protecting foreign investments from uncompensated seizure is one of the primary objectives of international investment agreements (IIAs).¹ At the same time, States have the right, and in many situations even the duty (eg under international human rights law), to regulate in the public interest.² Doing so may encroach upon proprietary rights of foreign investors. As pointed out in an UNCTAD study, 'almost any governmental measure could

¹ J-A Crawford and B Kotschwar, 'Investment' in A Mattoo, N Rocha and M Ruta (eds), *Handbook of Deep Trade Agreements* (World Bank Group 2020) 151, 161.

[ICLQ vol 71, October 2022 pp 945–974]

doi:10.1017/S0020589322000343

^{*} Associate Professor of International Economic Law, University of Canterbury, Aotearoa New Zealand, cnz. The author would like to thank the two anonymous reviewers for their very helpful comments.

² See eg Marvin Roy Feldman Karpa v United Mexican States, ICSID Case No ARB(AF)/99/1, Award (16 December 2002) para 103; S Faccio, 'Indirect Expropriation in International Investment Law: Between State Regulatory Powers and Investor Protection' (Editoriale Scientifica 2020) 129.

be construed as an act of interference in the business of a foreign investor'.³ Differentiating expropriatory measures from regulatory ones is one of the most vexing issues of international investment law. There is an underlying tension between private and public interests, between individual entitlements and national sovereignty, which is also germane to an understanding of the fair and equitable treatment (FET) standard.

It is important to stress at the outset that no IIA prohibits expropriation per se.⁴ It is lawful to expropriate subject to certain conditions. Those conditions are: (i) pursuance of a 'public purpose'; (ii) 'in a non-discriminatory manner'; (iii) 'on payment of ... compensation'; and (iv) under 'due process of law'.⁵ The conditions are cumulative ('and').⁶ That is, a lawful expropriation presupposes compensation *in addition to* public purpose, non-discrimination, and due process.⁷

At first blush, this only relates to the legality of expropriation; it does not bear upon its existence. States have sought to clarify the definition of expropriation in modern IIAs. To that effect, annexes on expropriation set out criteria to determine what amounts to expropriation and criteria the presence of which militates against measures amounting to expropriation. The *ratio legis* of the annexes is to provide tribunals with more guidance, or more to the point, to restrict their discretion when finding that expropriation has taken place, especially indirect expropriation. Accordingly, the provisions on expropriation 'shall be interpreted in accordance with' the relevant annexes.

³ UNCTAD, Taking of Property: UNCTAD Series on Issues in International Investment Agreements (2000) 6.

⁴ See also A Rajput, Regulatory Freedom and Indirect Expropriation in Investment Arbitration (Wolters Kluwer 2019) 7.

⁵ See eg art 8.12.1 Comprehensive Economic and Trade Agreement Between Canada, of the One Part, and the European Union and its Member States, of the Other Part (signed 30 October 2016, provisionally applied since 21 September 2017) (CETA); art 9.8.1 Trans-Pacific Partnership Agreement (TPP) as incorporated into the Comprehensive and Progressive Agreement for Trans-Pacific Partnership (signed 8 March 2018, entered into force 30 December 2018) (CPTPP) by virtue of art 1(1) thereof; art 14.8.1 United States-Mexico-Canada Agreement (signed 30 November 2018, entered into force 1 July 2020) (USMCA); art 10.13.1 Regional Comprehensive Economic Partnership Agreement (signed 15 November 2020, entered into force 1 January 2022) (RCEP).

⁶ A Reinisch and C Schreuer, *International Protection of Investments: The Substantive Standards* (CUP 2020) 189 para 898. See also *Bernardus Henricus Funnekotter v Republic of Zimbabwe*, ICSID Case No ARB/05/6, Award (22 April 2009) para 98.

⁷ Reinisch and Schreuer (n 6) 242 para 1182.

⁸ cf Annex 8-A to CETA; Annex 9-B to the TPP; Annex 14-B to the USMCA; Annex 10B to RCEP.

⁹ For CETA, see para 6(c) Joint Interpretative Instrument. J Kammerhofer, *International Investment Law and Legal Theory: Expropriation and the Fragmentation of Sources* (CUP 2021) 255; G Ünüvar, 'Is CETA the Promised Breakthrough? Interpretation and Evolution of Fair and Equitable Treatment and (Indirect) Expropriation Provisions' in M Andenas et al (eds), *EU External Action in International Economic Law* (Springer 2020) 213.

¹⁰ Art 8.12.1 CETA; fn 16 to art 9.8 TPP; art 14.8.5, fn 7 USMCA; fn 25 to art 10.13 RCEP. For the legal implications of this regulatory technique, see Kammerhofer (n 9) 257.

But the interplay of those annexes with the provisions on expropriation remains unclear.

Factors to be taken into account by tribunals are: the economic impact of the measure at issue; the extent of the government interference; the character of the measure (ie its 'objective' and 'object, context and intent');¹¹ and under the EU-Canada Comprehensive Economic and Trade Agreement (CETA), the duration of the measure. These factors are cumulative. 12 Ordinarily, international dispute settlement bodies are reluctant to probe States' intent. 13 That is why the Appellate Body in Japan—Alcoholic Beverages II, for instance, rejected an aim-and-effect test in the context of the national treatment (NT) obligation in Article III of the General Agreement on Tariffs and Trade (GATT). 14 In the words of the Appellate Body, 'It is irrelevant that protectionism was not an intended objective if the ... measure in question is nevertheless ... "applied to imported or domestic products so as to afford protection to domestic production".'15 This differs from the position concerning indirect expropriation, as 'intent' is explicitly listed as a relevant factor in CETA and United States-Mexico-Canada Agreement (USMCA). Another clarification—the fact that the measure 'has an adverse effect on the economic value of an investment ... does not establish that an indirect expropriation has occurred'16—rules out the application of the sole effect doctrine, according to which the effect of the measure at issue is the key criterion to determine expropriation, and is helpful in light of awards such as Metalclad v Mexico. 17 Still, tribunals have considerable discretion, first, because the lists of factors are non-exhaustive ('among other factors'), and secondly, in deciding whether the factors are present. 18

All this forms part of a wider attempt by States to properly calibrate investment protection.¹⁹ Whether States have achieved their goal as far as indirect expropriation is concerned will be examined in this article. In doing so, it will concentrate on the provisions concerning expropriation found in the following mega-regional trade and investment agreements: CETA, the Comprehensive and Progressive Agreement for Trans-Pacific Partnership

¹¹ Para 3(c) Annex 10B to RCEP; para 2(d) Annex 8-A to CETA; para 3(a)(iii) Annex 14-B to the USMCA.

¹² Contra Faccio (n 2) 106, but see the conjunction 'and' between the penultimate and ultimate factor listed, respectively.
13 cf ibid 155.

¹⁴ Appellate Body Report, *Japan—Taxes on Alcoholic Beverages*, WT/DS8/AB/R, WT/DS10/AB/R, WT/DS11/AB/R (1 November 1996) 27–8. ¹⁵ ibid 28 (emphasis in original). ¹⁶ Para 2(a) Annex 8-A to CETA; para 3(a)(i) Annex 9-B to the TPP; para 3(a)(i) Annex 14-B to

Para 2(a) Annex 8-A to CETA; para 3(a)(i) Annex 9-B to the TPP; para 3(a)(i) Annex 14-B to the USMCA; para 3(a) Annex 10B to RCEP.

Metalclad Corporation v United Mexican States, ICSID Case No ARB(AF)/97/1, Award (30 August 2000) para 103. The award was partially set aside in *The United Mexican States v Metalclad Corporation*, 2001 BCSC 664, Decision of the Supreme Court of British Columbia (2 May 2001). For the different doctrines to determine indirect expropriation, see M Malakotipour, 'The Chilling Effect of Indirect Expropriation Clauses on Host States' Public Policies: A Call for a Legislative Response' (2020) 22 ICLR 235, 238–42.

The more concrete formulation of the FET standard is another example.

(CPTPP),²⁰ the USMCA, and the Regional Comprehensive Economic Partnership Agreement (RCEP).²¹ Because of their size and the number of participating countries, mega-regionals set trends. It should be noted that an investor–State dispute settlement (ISDS) mechanism does not yet exist under RCEP (but is part of its work programme).²² Barring covered government contracts,²³ indirect expropriation is not actionable under the USMCA,²⁴ thus only leaving State–State dispute settlement.²⁵ Also, neither the CETA provision on expropriation nor its investor–State dispute resolution mechanism are operational yet.²⁶ Although this article focuses upon mega-regionals, similar provisions are also employed in bilateral investment treaties (BITs), notably treaties based upon the India Model BIT 2015.²⁷

The provisions that remove non-discriminatory regulatory measures from the definition of indirect expropriation have major implications for the domestic legal order. For example, paragraph 3(b) of Annex 14-B to the USMCA provides that '[n]on-discriminatory regulatory actions by a Party that are designed and applied to protect legitimate public welfare objectives ... do not constitute indirect expropriations, except in rare circumstances'. The Annex distinguishes regulatory actions from indirect expropriation and thus restricts the scope of the latter. States do not have to pay compensation to foreign investors in respect of such measures because they are not expropriatory, ²⁹ whereas for expropriatory measures compensation is a requirement of their legality. ³⁰

While the agreements under consideration all use similar wording, there are differences. CETA, the CPTPP and the USMCA make allowances for 'rare circumstances' whereas RCEP carves out non-discriminatory regulatory measures ipso facto. This raises the question of whether States were merely codifying existing case law along the lines of *Burlington v Ecuador*—where

²⁰ The CPTPP is in force for Australia, Canada, Japan, Mexico, New Zealand, Singapore, and Vietnam.

²¹ RCEP is in force for Australia, Brunei Darussalam, Cambodia, China, Japan, Lao PDR, New Zealand, Singapore, Thailand, and Vietnam.
²² Art 10.18.1(a) RCEP.

²³ See Annex 14-E to the USMCA.

²⁴ There is no ISDS mechanism between the United States and Canada under the USMCA, cf para 3 Annex 14-C thereof. For the non-actionability of indirect expropriation in the US–Mexico relationship, see art 14.D.3.1(a)(i)(B) and (b)(i)(B). The Canada–Mexico relationship is covered by the CPTPP.

²⁶ cf art 1(1)(a) Council Decision (EU) 2017/38 on the Provisional Application of the Comprehensive Economic and Trade Agreement (CETA) Between Canada, of the One Part, and the European Union and its Member States, of the Other Part, 2017 OJ L 11/1080.

Art 5.5 Bilateral Investment Treaty Between the Government of the Kyrgyz Republic and the Government of the Republic of India (signed 14 June 2019, not in force); art 5.5 Treaty Between the Republic of Belarus and the Republic of India on Investments (signed 24 September 2018, not in force).

^{28'} See eg para 3 Annex 8-A to CETA; para 3(b) Annex 9-B to the TPP; para 3(b) Annex 14-B to the USMCA; para 4 Annex 10B to RCEP.

²⁹ Faccio (n 2) 102. See also Rajput (n 4) 8, 195.

³⁰ Art 8.12.1(d) CETA; art 9.8.1(c) TPP; art 14.8.1(c) USMCA; art 10.13.1(c) RCEP.

the tribunal held that measures justified under the police powers doctrine are not expropriatory³¹—or whether they actually deviated from it: does the carve-out provide domestic regulators with more or less policy space than the indirect expropriation standards established by case law?

The article will proceed as follows: Part II explores the legal nature of the last paragraph of the annexes on expropriation and draws inferences from them concerning the steps of the legal analysis. Part III considers the legal requirements of the carve-out and the resulting level of scrutiny of domestic law, drawing upon a comparison with the law of the World Trade Organization (WTO). The key question addressed is whether host States can increase the level of protection given to legitimate public welfare objectives without incurring liability for expropriation. Part IV examines the 'rare circumstances' caveat, while Part V considers the burden of proof under the carve-out. Part VI looks at the question of compensation for regulatory measures in depth, with a particular focus on the ramifications of the already infamous ruling in *Eco Oro v Colombia*. ³² Part VII concludes.

II. THE LEGAL NATURE OF THE POLICE POWERS DOCTRINE AND ITS IMPLICATIONS

The last paragraph of the annexes on expropriation, such as paragraph 3(b) of Annex 14-B to the USMCA, embodies the police powers doctrine.³³ What is the legal nature of that last paragraph, which according to some has consequences for the distribution of the burden of proof?³⁴ Property is a *fictio iuris*; it is normative and needs to be fleshed out by law.³⁵ It is fair to say that international investment law impacts upon the concept of property not only through the definition of investment but also through its protection standards. Domestic property rights are protected internationally provided they constitute an investment, a threshold requirement under IIAs.

Furthermore, property is not an absolute right.³⁶ For instance, it is undisputed that proprietors may have to pay taxes on their property, such as real estate taxes.³⁷ The crux of the matter is determining when delineating the

³² Eco Oro Minerals v Republic of Colombia, ICSID Case No ARB/16/41, Decision on Jurisdiction, Liability and Directions on Quantum (9 September 2021).

Change Regulations Compensable? (2020) 35 ICSIDRev 154, 161, 173.

37 See eg art 1(2) Protocol to the Convention for the Protection

³¹ Burlington Resources v Republic of Ecuador, ICSID Case No ARB/08/5, Decision on Liability (14 December 2012) para 473.

³³ Eco Oro v Colombia (n 32) paras 625–6; Philip Morris Brand (Switzerland), Philip Morris Products (Switzerland) and Abal Hermanos (Uruguay) v Oriental Republic of Uruguay, ICSID Case No ARB/10/7, Award (8 July 2016) paras 300–301. See also UNCTAD, Expropriation: UNCTAD Series on Issues in International Investment Agreements II (2012) 88. For the origin of the police powers doctrine, see Rajput (n 4) 9–11.

of fin 18 to para 1 Annex 14-B to the USMCA; fin 1 to para 1 Annex 10B to RCEP.
 D Khachvani, 'Non-compensable Regulation Versus Regulatory Expropriation: Are Climate

³⁷ See eg art 1(2) Protocol to the Convention for the Protection of Human Rights and Fundamental Freedoms (done 20 March 1952, entered into force 18 May 1954) 213 UNTS 262. See also J Crawford, *Brownlie's Principles of Public International Law* (9th edn, OUP 2019) 604.

boundaries of property is tantamount to an encroachment. A domestic law which the investor considers as amounting to government interference, from the perspective of the host State, might amount to the mere delineation (regulation) of property rights, for example, the right to use the investment.

The way the last paragraphs of the annexes are formulated implies that they form part of the definition of indirect expropriation ('do not constitute indirect expropriations'). As a result, they need to be analysed when determining whether an expropriation has taken place.³⁸ They are consequently not exception clauses,³⁹ but limit the scope of what amounts to an expropriation and, as such, constitute exemptions or carve-outs.⁴⁰ The differentiation between exemptions and exceptions is also known from WTO law.⁴¹ According to the Panel in *EC—Approval and Marketing of Biotech Products*, 'the term "exemption" connotes freedom from, and hence the inapplicability of, an obligation'.⁴²

Any legal analysis must start with the various agreements' respective annex on expropriation, including the exemption. ⁴³ It is only after a finding has been made that a measure is expropriatory in accordance with the elements set out in the annex that the issue of legality, covered in the main body of their investment chapters, arises. If there is no expropriation, the question of its legality (and compensation) is irrelevant. As emphasized by UNCTAD, 'When the expropriatory nature of the measure is being opposed, it cannot be expected that the host State makes a pre-emptive payment.' ⁴⁴ In line with this, the arbitral tribunal in *Allard v Barbados* noted that 'the first requirement for a successful claim of indirect expropriation is that ... the investor [should] have suffered a substantial deprivation of the use or expected economic benefit of the investment'. ⁴⁵ The implications of some of the legality requirements also being definitional elements will be discussed later. ⁴⁶

It can be inferred from the exemptions that regulatory action may amount to indirect expropriation since the definitional carve-out only relates to regulatory

³⁸ *Pro* SW Schill and HL Bray, 'The Brave New (American) World of International Investment Law: Substantive Investment Protection Standards in Mega-Regionals' in T Rensmann (ed), *Mega-Regional Trade Agreements* (Springer 2017) 139.

In this sense, Kammerhofer (n 9) 250; see also 239.

⁴⁰ Pro Crawford and Kotschwar (n 1) 153, 162. See also Ünüvar (n 9) 211; JE Viñuales, 'Seven Ways of Escaping a Rule: Of Exceptions and Their Avatars in International Law' in F Paddeu and L Bartels (eds), Exceptions in International Law (OUP 2020) 68. For the concept of 'scope limitations', see C Henckels, 'Scope Limitation or Affirmative Defence? The Purpose and Role of Investment Treaty Exception Clauses' in F Paddeu and L Bartels (eds), Exceptions in International Law (OUP 2020) 364–5.

See eg Panel Report, European Communities—Measures Affecting the Approval and Marketing of Biotech Products, WT/DS291/R, WT/DS292/R, WT/DS293/R (21 November 2006) paras 7.2972, 7.2997.
 ibid, para 7.2972 (reference omitted).
 For the sequence of analysis and the different approaches thereto, see Eco Oro v Colombia (n 32)

paras 627–9.

44 UNCTAD, Expropriation (n 33) 43.

45 Allard v Barbados, PCA Case No 2012-06, Award (27 June 2016) para 263 (emphasis

⁴⁵ Allard v Barbados, PCA Case No 2012-06, Award (27 June 2016) para 263 (emphasis added). See also Windstream Energy v Government of Canada, PCA Case No 2013-22, Award (27 September 2016) para 285.

⁴⁶ See below Pt III.B.

action that is 'non-discriminatory' and 'designed and applied to protect legitimate public welfare objectives'. These cumulative⁴⁷ elements will be considered in the next section.

III. ELEMENTS OF THE EXEMPTIONS

A. Regulatory Actions

Before considering what is meant by the 'non-discriminatory' element and the 'designed and applied to protect' test, a word on 'regulatory actions' is in order. Whereas CETA merely speaks of 'non-discriminatory measures', the CPTPP, the USMCA and RCEP refer to 'non-discriminatory *regulatory* actions'.⁴⁸ The omission in CETA does not suggest a different scope, just as there is no difference in meaning between 'measures' and 'actions', both referencing acts of State within the meaning of the International Law Commission's Articles on State Responsibility.⁴⁹ One commentator raises the question of 'whether the term "regulatory measures" refer[s] only to generally applicable laws and regulations, or whether it encompasses both general measures and specific measures targeted at the investor'.⁵⁰ This question will be discussed in the next section on non-discrimination.

B. Non-Discrimination

For a measure to be exempt under the carve-outs, the measure must be non-discriminatory. The exemptions thus combine the concept of indirect expropriation with the concept of non-discrimination. Discriminatory State conduct is also captured by other provisions in IIAs. The most-favoured-nation (MFN) and NT obligations are concerned with nationality-based discrimination. A Drafters' Note to Articles 9.4 and 9.5 of the TPP confirms that these provisions 'seek to ensure that foreign investors or their investments are not treated less favourably *on the basis of their nationality*'. Non-discrimination as part of the FET standard, by contrast, is wider and includes discrimination on grounds 'such as gender, race or religious belief'. Sa

There appears to be an inherent tension between the elements of the exemptions, on the one hand, and the requirements of a lawful expropriation, on the other hand.⁵⁴ This is because non-discrimination and the pursuance of

 ⁴⁷ Faccio (n 2) 113.
 ⁴⁸ Emphasis added.
 ⁴⁹ International Law Commission (ILC) Draft Articles on Responsibility of States for Internationally Wrongful Acts, GAOR 56th Session Supp 10, 43 (2001) (Articles on State Responsibility). For 'measures' in general, see Rajput (n 4) 8–9.
 ⁵⁰ Ünüvar (n 9) 211.

cf arts 8.6–8.7 CETA; arts 9.4–9.5 TPP; arts 14.4–14.5 USMCA; arts 10.3–10.4 RCEP.
 Para 2, 3rd sentence (emphasis added).
 See eg art 8.10.2(d) CETA.

⁵⁴ cf I Prezas, 'Équation insoluble ? L'annexe interprétative du CETA relative à l'expropriation indirecte à l'épreuve de la technique européenne de proportionnalité' in C Titi (ed), *Droits de*

legitimate public welfare objectives are both prerequisites of a lawful expropriation. ⁵⁵ Article 14.8.5 of the USMCA stipulates that 'whether an action or series of actions by a Party *constitutes* an expropriation shall be determined in accordance with paragraph 1 of this Article *and* Annex 14-B (Expropriation)', ⁵⁶ which seems to suggest that the legality requirements are also germane to a finding of expropriation. The requirement of 'public purpose' would be met if the measure at issue protected a public welfare objective in terms of the Annex. Both requirements are equally open-ended and, therefore, equivalent. That said, their interchangeability is not undisputed. ⁵⁷ It is suggested here that whilst 'public welfare objectives' is the modern concept, drafters continue to use the term 'public purpose' in the provisions on expropriation for customary reasons. ⁵⁸

This duplication of requirements creates friction between the annexes and the provisions on expropriation: on the one hand, it is said that non-discriminatory regulatory measures are non-compensable (except in rare circumstances), whilst the provisions on expropriation make compensation a requirement if they are to qualify as non-discriminatory *expropriatory* measures for a public purpose. Put differently, as with the provisions on expropriation, compensation is required in the case of expropriatory measures which are for a public purpose and non-discriminatory. Under the carve-outs, regulatory measures are exempt if 'designed and applied to protect legitimate public welfare objectives'—that is, for a public purpose—and non-discriminatory. Assuming there to have been due process, there appears to be no difference between both categories of measures—compensable expropriatory and non-compensable regulatory measures—except for the premise that one category is expropriatory and the other is regulatory in nature. ⁵⁹ This calls the added value of the exemptions into question.

Conflicting case law illustrates the dilemma. Although it concerns direct expropriation, a good starting point is the *Santa Elena v Costa Rica* case. There, the tribunal highlighted the fact that because 'public purpose' is a legality requirement, on its own it is insufficient to demarcate compensable expropriation from non-compensable regulatory action: 'where property is expropriated, even for environmental purposes ..., the state's obligation to pay compensation remains'.⁶⁰ This pronouncement assumes expropriation.

l'homme et droit international économique (Bruylant 2019) 168–72; Faccio (n 2) 111. See generally Kammerhofer (n 9) 228–49.

See also UNCTAD, *Expropriation* (n 33) 95.

⁵⁶ Emphasis added.

⁵⁷ Instead of all, see Y Radi, Rules and Practices of International Investment Law and Arbitration (CUP 2020) 175–8.

⁵⁸ cf fin 17 to art 9.8.1(a) TPP. See also *Amoco International Finance Corporation v Iran*, Iran-US Claims Tribunal, Partial Award (14 July 1987) para 115; R Dolzer and C Schreuer, *Principles of International Investment Law* (2nd edn, OUP 2012) 99.

⁵⁹ See also Kammerhofer (n 9) 241.

⁶⁰ Compañia del Desarrollo de Santa Elena v Costa Rica, ICSID Case No ARB/96/1, Award, as rectified by Award of 8 June 2000 (17 February 2000) para 72.

We are concerned, however, with whether there has actually been an expropriation, which logically precedes the question of legality.⁶¹ This is when the police powers doctrine comes into play.

The OECD defined that doctrine as 'an accepted principle of customary international law that where economic injury results from a bona fide non-discriminatory regulation ... compensation is not required'. ⁶² The doctrine's reach is contested. On the one hand, the arbitral tribunal in *Methanex v USA* held that 'a non-discriminatory regulation for a public purpose, which is enacted in accordance with due process and, which affects, inter alios, a foreign investor or investment is not deemed expropriatory and compensable ...'. ⁶³ On the other hand, given how rare direct expropriations are in practice, ⁶⁴ the tribunal in *Pope & Talbot* noted that 'a blanket exception for regulatory measures would create a gaping loophole in international protections against expropriation.' ⁶⁵ By the same token, the tribunal in *Vivendi v Argentina (II)* observed that 'If public purpose automatically immunises the measure from being found to be expropriatory, then there would never be a compensable taking for a public purpose.' ⁶⁶

The tribunal in *Saluka v Czech Republic* steered a middle course and ruled that 'States are not liable to pay compensation to a foreign investor when, *in the normal exercise of their regulatory powers*, they adopt in a non-discriminatory manner *bona fide* regulations that are aimed at the general welfare.'⁶⁷ This, of course, begs the question of what constitutes the 'normal' exercise of regulatory powers. In a similar vein, the arbitral tribunal in *Quiborax v Bolivia* opined that 'the taking must not qualify as the *legitimate exercise* of the State's police powers'.⁶⁸ Case law has tried to reify 'normal'/'legitimate' in this context,⁶⁹ and the exemptions have codified that case law:⁷⁰ the exercise of police powers is legitimate when it is, first, non-discriminatory, and secondly, 'designed and applied to protect legitimate public welfare objectives'.

⁶¹ See above Pt II.

⁶² OECD, "Indirect Expropriation" and the "Right to Regulate" in International Investment Law' (September 2004) OECD Working Papers on International Investment 2004/04, 5 fn 10.

Methanex Corporation v United States of America, UNCITRAL, Final Award on Jurisdiction and Merits (3 August 2005) Pt IV, Ch D, para 7.
Pope & Talbot v Government of Canada, UNCITRAL, Interim Award (26 June 2000)

para 99.

66 Compañía de Aguas del Aconquija and Vivendi Universal v Argentine Republic (II), ICSID

Case No ARB/97/3, Award (20 August 2007) para 7.5.21.

67 Saluka Investments v Czech Republic, UNCITRAL, Partial Award (17 March 2006) para 255

⁽emphasis added).

68 Quiborax and Non-metallic Minerals v Plurinational State of Bolivia, ICSID Case No ARB/06/2, Award (16 September 2015) para 200 (emphasis added).

⁶⁹ See eg Mohamed Abdel Raouf Bahgat v Arab Republic of Egypt (I), PCA Case No 2012-07, Final Award (23 December 2019) paras 230–32; Philip Morris v Uruguay (n 33) paras 305–7; Técnicas Medioambientales Tecmed v United Mexican States, ICSID Case No ARB(AF)/00/2, Award (29 May 2003) paras 121–2.

⁷⁰ See also F Baetens, 'Protecting Foreign Investment and Public Health Through Arbitral Balancing and Treaty Design' (2022) 71 ICLQ 139, 167.

How is sense to be made of this? If there was no difference between the legality requirements of expropriation and the requirements of the exemptions, this would mean that the exemptions are little more than guidelines for treaty interpreters to the effect that, under *normal* circumstances, regulatory actions are not compensable.⁷¹ Hence, only the 'rare circumstances' caveat would have practical relevance, which would also entail that the RCEP exemption—by omitting that caveat—is inconsequential, as it could not resolve the internal conflict noted above.

The solution is to distinguish between non-discrimination as a requirement of the legality of expropriation and as a requirement of the exemptions. As to the legality requirement, the arbitral tribunal in *Teinver v Argentina* found that 'Discrimination requires differential treatment of Claimants' investment from other similar investments in like circumstances.'⁷² To find discrimination in this connection, something more is required than the simple designation of the investment by the host State for expropriation. An expropriatory act, such as a seizure, will always be addressed to, and in that sense target, an investor. This alone would not make the act unlawful. The salient point is whether the host State would have acted differently if the investor was someone else.

It should be noted that, in the context of the exemptions, 'non-discriminatory' does not assume a rational relationship between the measure at issue and the public welfare objective pursued. As will be seen later, this is covered by the 'designed to protect' element. Returning to the OECD definition, it defined non-discrimination as part of the police powers doctrine as follows: 'A state measure will be discriminatory if it results "in an actual injury to the alien ... with the intention to harm the aggrieved alien" to favour national companies. However, this nationality-based definition does not add anything to the MFN/NT obligations, nor to the legality requirement for expropriation: an MFN/NT breach would automatically make the taking of property unlawful, because 'No Party shall expropriate ..., expect ... in a non-discriminatory manner'. To

The exemptions hark back to Article 11(a)(ii) of the Convention Establishing the Multilateral Investment Guarantee Agency, the relevant part of which reads as follows: 'with the exception of non-discriminatory measures of general application which governments normally take for the purpose of regulating economic activity in their territories'. The accompanying commentary defines 'non-discriminatory measures of general application' as including 'taxation, environmental and labor legislation as well as normal measures for

⁷¹ In this sense, Prezas (n 54) 172.

⁷² Autobuses Urbanos del Sur, Teinver and Transportes de Cercanías v Argentine Republic, ICSID Case No ARB/09/1, Award (21 July 2017) para 1019.
73 See below Pt III.C.1.a)(1).
74 OECD (n 62) 5 fn 10.

⁷⁵ Art 9.8.1(b) TPP; art 14.8.1(b) USMCA; art 10.13.1(b) RCEP; see also art 8.12.1(c) CETA.
76 Convention Establishing the Multilateral Investment Guarantee Agency (done 11 October 1985, entered into force 12 April 1988) (1985) 24 ILM 1598.

the maintenance of public safety'. 77 The 'general application' part was omitted in the exemptions, which suggests that it was deemed superfluous by the drafters in light of the non-discrimination requirement. The CPTPP adds to that list of measures of general application 'competition, ... health or other regulatory laws'.78

Against this backdrop, the reference to 'non-discriminatory' measures in the exemptions, and by extension the definition of indirect expropriation, appears to make a distinction between measures of general application and measures that single out the claimant investor. ⁷⁹ In Eco Oro v Colombia, the tribunal deduced from the fact that 'The measures affected all mining concessionaires' that they were non-discriminatory.⁸⁰ To be able to bring a claim under CETA, the CPTPP, or the USMCA, the investor must have suffered 'loss or damage' by reason of the alleged breach, making the incurrence of loss or damage an admissibility requirement.⁸¹ This has an impact on this. If others, be they nationals or foreigners, suffer the same loss or damage as a result of the measure at issue, this is an indication that the measure is of general application, in other words, that it constitutes a regulatory measure.⁸² By understanding the term 'discriminatory' here as meaning that the measure is not of general application—which is different from the investment protection standards where it refers to differential treatment on grounds of nationality, gender, race, etc—the principle of effectiveness is respected.⁸³

In summary, non-discrimination in the context of exemptions does not refer to nationality-based differential treatment, as in the MFN and NT obligations as well as the legality requirement for expropriation itself.⁸⁴ Rather, it implies that the measure at issue is of general application, understood as meaning the 'nonselective application of laws and regulations'. 85 This means that such measures were not designed to apply specifically to the claimant investor. The fact that, for example, the revocation of an operating licence may be non-compensable does not militate against this finding, as the underlying law upon which the revocation was based might itself be of general application. That some IIAs have no MFN and NT obligations does not change the meaning of non-

⁷⁷ Multilateral Investment Guarantee Agency, 'Commentary on the Convention Establishing

⁸⁰ Eco Oro v Colombia (n 32) paras 640–41.

⁸¹ Art 8.18.1 CETA; art 9.19.1(a)(ii) and (b)(ii) TPP; para 2(a)(ii) and (b)(ii) Annex 14-E to the USMCA concerning covered government contracts. As mentioned before, RCEP does not have an ISDS mechanism, and the CETA investor-State dispute resolution mechanism is not in operation For regulatory measures, see also art 8.9 CETA; art 9.16 TPP; art 14.16 USMCA.

Instead of all, see Appellate Body Report, United States-Standards for Reformulated and Conventional Gasoline, WT/DS2/AB/R (20 May 1996) 23 ('interpretation must give meaning and effect to all the terms of a treaty').

⁸⁴ For the legality requirement, see eg ADC Affiliate and ADC & ADMC Management v Republic of Hungary, ICSID Case No ARB/03/16, Award (2 October 2006) para 442. 85 Khachvani (n 36) 161.

discrimination in the exemptions either, because the omission of those obligations is to be respected as reflecting the common intention of the contracting parties.⁸⁶

The next section will consider the 'designed and applied to protect' standard, including the proportionality test which it embodies and which is key to determining whether a measure of general application is non-compensable under the exemptions. It is key because it determines the level of scrutiny of, or, in other words, the level of intrusion into, domestic law.

As for the question of 'objectives', the arbitral tribunal in *Oxus Gold v Uzbekistan* observed that 'in case of indirect expropriation, there is a State law or regulation, or sometimes some behaviour, – the purpose of which was not to take the property'.⁸⁷ Just like the legality requirement of 'public purpose', a public welfare objective's legitimacy (and hence its protectability in principle) will rarely be challenged by claimant investors, bearing in mind that the list of public welfare objectives is non-exhaustive ('such as').⁸⁸ The Preambles, which form part of the relevant context within the meaning of Article 31(2) of the Vienna Convention on the Law of Treaties (VCLT),⁸⁹ expand that list further.⁹⁰ This allows host States to pursue multiple public welfare objectives with the same measure.⁹¹

C. Designed and Applied to Protect

All substantive investment protection standards subject domestic law to international scrutiny. The level of scrutiny by investment tribunals depends upon the type of proportionality test embodied in the 'designed and applied' standard. Suitability, necessity and strict proportionality need to be considered. ⁹² It should be emphasized that the issue here relates to whether an indirect expropriation has taken place, not whether the amount of compensation is proportionate.

⁸⁶ cf Methanex v USA (n 63) Pt IV, Ch C, para 25. See also Rajput (n 4) 176.

⁸⁷ Oxus Gold v Uzbekistan, UNCITRAL, Final Award (17 December 2015) para 740.

⁸⁸ cf CL Lim, J Ho and M Paparinskis, *International Investment Law and Arbitration:* Commentary, Awards and Other Materials (CUP 2018) 338.

⁸⁹ Vienna Convention on the Law of Treaties (concluded 23 May 1969, entered into force 27 January 1980) 1155 UNTS 331.

Para 6 Preamble to CETA; para 6 Preamble to the CPTPP and para 9 Preamble to the TPP;
 para 9 Preamble to the USMCA.
 T Meyer, 'A Political Theory of Legal Exceptions' (2021) Vanderbilt Law Research Paper No

⁹¹ T Meyer, 'A Political Theory of Legal Exceptions' (2021) Vanderbilt Law Research Paper No 21-18, 67.

⁹² See also J Paine, 'Autonomy to Set the Level of Regulatory Protection in International Investment Law' (2021) 70 ICLQ 697, 700, 726, 729–30; Rajput (n 4) 185; F Ortino, 'Defining Indirect Expropriation: The Transatlantic Trade and Investment Partnership and the (Elusive) Search for "Greater Certainty" (2016) 43 LIEI 351, 363. For proportionality in general, see R Bradshaw, 'Legal Stability and Legitimate Expectations: Does International Investment Law Need a Sense of Proportion?' (2020) 5 EILAR 240, 242–50.

The arbitral tribunal in *Electrabel v Hungary* defined proportionality as requiring 'the measure to be suitable to achieve a legitimate policy objective, necessary for that objective, and not excessive considering the relative weight of each interest involved'. 93 According to De Brabandere and Baldini Miranda da Cruz, 'Proportionality is a general principle of international law that presupposes that an examination of a possible conflict between different rights will follow some general rational rules that have the goal of ensuring that even restricted rights will not be completely suppressed.'94 Both in domestic and international law, the principle, or technique, 95 of proportionality helps settle conflicts between competing rights and interests in a systematic manner. 96 Often, such conflicts are at the root of an investment dispute: the expansion of a mining project within First Nation land affects Indigenous rights; the construction of a hotel complex next to a national park affects the environment; conversely, climate change measures imposing emissions reductions affect investments in the energy and primary industries sectors.⁹⁷ Claimants regularly challenge the policy choices and prioritization decisions made by host States under the heads of FET and indirect expropriation.

It has long been recognized that proportionality is instrumental in determining whether there is indirect expropriation.⁹⁸ The crux of the matter is what is or is not proportionate.⁹⁹ This hinges, in particular, upon the meaning of 'designed ... to protect'. A suitability test would be a low threshold.¹⁰⁰ Bradshaw summarizes that test as follows: 'In order to be

⁹³ Electrabel v Republic of Hungary, ICSID Case No ARB/07/19, Award (25 November 2015) para 179 (reference omitted); confirmed in Hydro Energy 1 and Hydroxana Sweden v Kingdom of Spain, ICSID Case No ARB/15/42, Decision on Jurisdiction, Liability and Directions on Quantum (9 March 2020) para 574.

⁹⁴ E De Brabendere and P Baldini Miranda da Cruz, 'The Role of Proportionality in International Investment Law and Arbitration: A System-specific Perspective' in U Linderfalk and E Gill-Pedro (eds), Revisiting Proportionality in International and European Law: Interests and Interest-holders (Brill 2021) 199.
⁹⁵ Prezas (n 54) 148.

⁹⁶ De Brabendere and Baldini Miranda da Cruz (n 94) 199, 202, 218.

For climate change measures, in particular, see Khachvani (n 36) 156, 169–72.

⁹⁸ See eg *Tecmed v Mexico* (n 69) para 122; *LG&E Energy, LG&E Capital and LG&E International v Argentine Republic*, ICSID Case No ARB/02/1, Decision on Liability (3 October 2006) para 194. See also para 3(b) Annex 13 Free Trade Agreement Between the Government of New Zealand and the Government of the People's Republic of China (signed 7 April 2008, entered into force 1 October 2008); para 3(c) Annex 2 to the Association of South-East Asian Nations Comprehensive Investment Agreement (signed 26 February 2009, entered into force 24 February 2012) (ASEAN CIA); paras 3(a)(iii), (b) Annex 10-A to the Comprehensive Economic Partnership Agreement Between India and the Republic of Korea (signed 7 August 2009, entered into force 1 January 2010) (India–Korea CEPA). See generally Reinisch and Schreuer (n 6) 168–70. Critical, C Foster, 'Why Due Regard Is More Approriate than Proportionality Testing in International Investment Law' (2022) 23 JWIT 388, 390, 394–96, 405–6, 410, 415–16 (primarily of proportionality sensu stricto); Kammerhofer (n 9) 242; Rajput (n 4) 189.

⁹⁹ De Brabendere and Baldini Miranda da Cruz (n 94) 218–19 argue that there is not one proportionality test in international investment law but many.

¹⁰⁰ RWE Innogy and RWE Innogy Aersa v Kingdom of Spain, ICSID Case No ARB/14/34, Decision on Jurisdiction, Liability and Certain Issues of Quantum (30 December 2019) para 554.

suitable, ... the measure must be capable of furthering the policy objective. This requires a rational connection between the desired end and the means (i.e. the regulatory measure) chosen to advance it.'101 Rajput notes that a suitability test is 'akin to the standard of reasonableness'.102

WTO jurisprudence has concretized the concept of necessity. It is comprised, inter alia, of a 'determination of whether a WTO-consistent alternative measure which the Member concerned could "reasonably be expected to employ" is available, or whether a less WTO-inconsistent measure is "reasonably available". This has been endorsed by investment tribunals. Wecessity in this context, as an element of proportionality, is not to be confused with a circumstance precluding wrongfulness in terms of Article 25 of the ILC's Articles on State Responsibility under which, in order to be deemed necessary, the measures adopted need to be 'the only way' for the State to protect the public welfare objective in question. 105

A strict proportionality test would be the most intrusive approach since it allows claimant investors to challenge the host State's decision to prioritize a particular public welfare objective. If successful, the investor's interest trumps the protected objective. An investor could, therefore, bring about a reduction of the level of protection offered to a public welfare objective in order to protect its own, competing, interest because, in the words of the arbitral tribunal in *PL Holdings v Poland*, the measure is 'excessive in that its advantages are outweighed by its disadvantages'. ¹⁰⁶ Under a strict proportionality test, the adjudicatory body weighs up 'the charge or weight imposed to the foreign investor and the aim sought to be realized by any expropriatory measure'. ¹⁰⁷

The 'designed and applied' test has two elements: the design element and the application element. The former concerns the measure itself, the latter relates to the way the measure has been implemented. These will be addressed separately. It should be noted that the 'designed and applied' test is not the only way in which non-discriminatory regulatory measures are carved out from the scope of indirect expropriation. Canada's 2021 Model Foreign Investment Promotion and Protection Agreement (FIPA) employs the following clause:

¹⁰¹ Bradshaw (n 92) 246.
103 Appellate Body Report, Korea—Measures Affecting Imports of Fresh, Chilled and Frozen Beef, WT/DS161/AB/R, WT/DS169/AB/R (10 January 2001) para 166. See also Appellate Body Report, European Communities—Measures Prohibiting the Importation and Marketing of Seal Products, WT/DS400/AB/R, WT/DS401/AB/R (18 June 2014) paras 5.169, 5.214; Appellate Body Report, Colombia—Measures Relating to the Importation of Textiles, Apparel and Footwear, WT/DS461/AB/R (22 June 2016) para 5.70 ('a comparison between the challenged measure and possible alternatives').

Contra Rajput (n 4) 185–6.

PL Holdings v Poland, SCC Case No 2014/163, Partial Award (28 June 2017) para 355. For European law, see ECJ, Case C-331/88 The Queen v Ministry of Agriculture, Fisheries and Food [1990] ECR I-4023, para 13 ('the disadvantages caused must not be disproportionate to the aims pursued'); Case of Handyside v The United Kingdom App No 5493/72 (ECtHR, 7 December 1976) para 49 ('proportionate to the legitimate aim pursued').

Tecmed v Mexico (n 69) para 122.

'A non-discriminatory measure ... that is adopted and maintained in good faith to protect legitimate public welfare objectives ... does not constitute indirect expropriation'108 So far, however, this language has not been employed in any agreement.¹⁰⁹

1. Designed to protect

The design element is indicative of a less strict proportionality test. A strict proportionality test would be irreconcilable with the much simpler 'designed and applied' test used in the exemptions and must be considered to have been superseded by that test. In 'rare circumstances', however, a stricter test may still apply. 110 The rejection of proportionality sensu stricto is also rooted in the fact that an investment tribunal is not a constitutional court and does not have the same powers. 111 This more deferential reading of investment protection standards is borne out by pronouncements of recent arbitral tribunals. For instance, the tribunal in *Muszynianka v Slovakia* found that 'it is not the role of an arbitral tribunal to "weigh the wisdom of legislation" ... arbitral tribunals must pay deference to the choices States make when deciding how to implement policy objectives'. 112

a) Suitability

'Designed to protect' implies, at the very least, a suitability test, this being the first step of proportionality. The measure at issue must pursue a legitimate public welfare objective. But this alone would not be sufficient to distinguish non-discriminatory regulatory action from indirect expropriation, because for an expropriation to be legal, it must be 'for a public purpose', in other words, pursue a legitimate public welfare objective. 113 Returning to the Santa Elena v Costa Rica case, the tribunal said that 'While an expropriation or taking for environmental reasons may be classified as a taking for a public purpose, and thus may be legitimate, the fact that the Property was taken for this reason does not affect either the nature or the measure of the compensation to be paid for the taking.'114 Henckels concludes that 'whether a direct expropriation has been effected for a public purpose is only relevant insofar as it affects the issue of whether an expropriation is lawful or unlawful, and not the issue of whether

Art 9.3, 2nd sentence.
 Paine (n 92) 730–31. See below Pt III.C.2.

¹¹¹ Foster (n 98) 410. For European Union law, see ECJ, Opinion 1/17, ECLI:EU:C:2019:341 (2019) paras 148, 156, 160.

Muszynianka v Slovak Republic, PCA Case No 2017-08, Award (7 October 2020) paras 557–8 (reference omitted). See also *Philip Morris v Uruguay* (n 33) para 399, citing *Electrabel v Republic of Hungary*, ICSID Case No ARB/07/19, Decision on Jurisdiction, Applicable Law and Liability (30 November 2012) para 8.35.

Compañia del Desarrollo de Santa Elena v Costa Rica (n 60) para 71.

an expropriation has taken place'. 115 For regulatory measures, things are different: whether or not the host State pursued a public purpose (ie a legitimate public welfare objective) also forms part of the definition of indirect expropriation, and hence is germane to the question of whether the measure at issue is expropriatory in the first place.

That said, the design element requires more than just the identification of the legitimate public welfare objective. On the spectrum of scrutiny, the more stringent step would be to consider whether the measure at issue is not incapable of promoting the protection of the stated objective. At this juncture, it is useful to canvass WTO law. Inferences may be drawn from international trade law because of the systemic function fulfilled by international investment law in the global economic system, which is complementary to international trade regulation. 116

(1) Means-end nexus

WTO law also has a design element. Under WTO law this is a preliminary to the necessity test. ¹¹⁷ The Appellate Body deduces this from the phrase 'to protect'. One could equally argue that necessity encompasses suitability. ¹¹⁸ In *Colombia—Textiles*, the Appellate Body observed with respect to Article XX(a) of the GATT that

the phrase 'to protect public morals' calls for an ... examination in order to determine whether there is a relationship between an otherwise GATT-inconsistent measure and the protection of public morals. If this ... examination reveals that the measure is *incapable of protecting* public morals, there is not a relationship between the measure and the protection of public morals that meets the requirements of the 'design' step. 119

The Appellate Body thus established that the design element involves an examination of the relationship between the measure at issue and the public welfare objective pursued. This is reminiscent of the 'relating to' test in Article XX(g) of the GATT, which requires 'a close and genuine relationship of ends and means'. The Panel in *Brazil—Taxation* sums up the case law as follows: the 'designed to protect' element amounts to an analysis of 'whether

¹¹⁵ C Henckels, Proportionality and Deference in Investor-State Arbitration: Balancing Investment Protection and Regulatory Autonomy (CUP 2015) 173 (emphasis added).

cf De Brabendere and Baldini Miranda da Cruz (n 94) 207. Contra Rajput (n 4) 186, 189.
 For art XX(a) GATT, see Appellate Body Report, Colombia—Textiles (n 103) para 5.67; for art XX(d) GATT, see Appellate Body Report, Korea—Various Measures on Beef (n 103) para 157.
 cf Appellate Body Report, EC—Seal Products (n 103) para 5.169.

Appellate Body Report, Colombia—Textiles (n 103) para 5.68 (reference omitted, emphasis added).

Appellate Body Report, China—Measures Related to the Exportation of Rare Earths, Tungsten and Molybdenum, WT/DS431/AB/R, WT/DS432/AB/R, WT/DS433/AB/R (29 August 2014) paras 5.90, 5.117; Appellate Body Report, China—Measures Related to the Exportation of Various Raw Materials, WT/DS394/AB/R, WT/DS395/AB/R, WT/DS398/AB/R (22 February

the measure is not incapable of contributing to that objective'. 122 According to the Appellate Body, this requires an assessment of the measure's 'content, structure, and expected operation', 123 This final step, the 'expected operation', is expressly covered by the 'applied to protect' requirement in the exemptions.

(2) Is the measure apt to make a contribution?

According to the Appellate Body, the design element is not 'a particularly demanding step'. 124 In WTO law, whether the measure makes, or will make, a contribution to the achievement of the stated public welfare objective is subjected to a test of necessity. 125 Therefore, under the exemptions, a respondent could argue that it does not matter if the measure at issue actually contributes to the protection of the policy objective as long as the measure was designed for that purpose. In other words, an adjudicatory body would not gauge the effectiveness of the measure, for even ineffective measures could be 'designed ... to protect ... public welfare objectives'. 126 Indeed, with respect to 'relating to', the Appellate Body confirmed in China—Rare Earths that the term 'does not require an examination of the actual effects of the challenged measure'.127

The problem with applying this approach to the exemptions is that the only requirement that a measure would have to fulfil would be the pursuance of a legitimate public welfare objective. Given that the list of such objectives is non-exhaustive, this is a high—and in the author's view inadequate standard of review, which accords the host State too much deference in the context of indirect expropriation.

Whether the measure is capable of making a contribution to the achievement of the public welfare objective, correctly understood, is a question of suitability and should be scrutinized as such. 128 This is buttressed by the way the Appellate Body defines that element: 'A contribution exists when there is a genuine relationship of ends and means between the objective pursued and the measure at issue.'129 It is true that the Appellate Body also stated that for the

²⁰¹²⁾ para 355; Appellate Body Report, United States—Import Prohibition of Certain Shrimp and Shrimp Products, WT/DS58/AB/R (6 November 1998) para 136.

Panel Report, Brazil—Certain Measures Concerning Taxation and Charges, WT/DS472/R, WT/DS497/R (11 January 2019) para 7.519.

Appellate Body Report, *Colombia—Textiles* (n 103) para 5.69. 124 ibid, para 5.70. Appellate Body Report, *Brazil—Measures Affecting Imports of Retreaded Tyres*, WT/DS332/ AB/R (17 December 2007) para 178. The Appellate Body clarified that a trade-restrictive measure, 'the contribution of which is not immediately observable', could still be justifiable, provided the measure 'is apt to produce a material contribution to the achievement of its objective', see para 151.

¹²⁶ Emphasis added.
127 Appellate Body Report, *China—Rare Earths* (n 121) paras 5.114, 5.118, 5.138, 5.147.

See also F Ortino, 'Investment Treaties, Sustainable Development and Reasonableness Review: A Case Against Strict Proportionality Balancing' (2017) 30 LJIL 71, 87-8.

Appellate Body Report, Brazil—Retreaded Tyres (n 125) para 210.

purposes of WTO law 'using a "contribution" test for the "relating to" analysis, is not, by itself, an appropriate substitute for a holistic assessment of whether a measure has a close, genuine, and substantial relationship to' a legitimate public welfare objective. ¹³⁰ At the same time, however, the Appellate Body found that 'consideration of the predictable effects of a measure may be relevant'. 131

Indeed, both 'relating to' and 'necessity' connote varying degrees of a means—end nexus, with the former being looser. 132 One of the main purposes of the annexes on expropriation is to clarify the means—end nexus required for regulatory measures, and hence the degree of proportionality required. Under German law, where the proportionality test originates, 133 the salient point under the suitability prong is whether the measure 'could further the desired outcome'. 134 From this it follows, that a measure is not suitable if it cannot further the desired outcome, or impedes its achievement. 135

That said, whether or not a measure makes a 'contribution' to achieving the objective is only one aspect of the necessity analysis. 136 Other factors are 'the importance of the objective', the 'restrictiveness of the measure' and, as mentioned above, 'a comparison between the challenged measure and possible alternatives should ... be undertaken'. 137 It is, therefore, unquestionable that the 'designed to protect' test gives regulators more leeway than necessity, ¹³⁸ which in addition requires those scrutinizing a measure 'to assess, in a qualitative or quantitative manner, the extent of the measure's contribution to the end pursued, rather than merely ascertaining whether or not the measure makes any contribution'. 139 This has the advantage of more easily accommodating measures that have more than one regulatory objective. 140

As far as scrutiny goes, the 'designed to protect' standard thus sits between a suitability and a necessity test. If the drafters had wanted a fully-fledged

¹³⁰ Appellate Body Report, *China—Rare Earths* (n 121) para 5.117.

ibid, paras 5.100, 5.113, 5.138, 5.147.

cf Appellate Body Report, US—Gasoline (n 83) 17–18; Appellate Body Report, US—Shrimp (n 121) para 141.

Bradshaw (n 92) 242; C Henckels, 'Indirect Expropriation and the Right to Regulate: Revisiting Proportionality Analysis and the Standard of Review in Investor-State Arbitration' (2012) 15 JIEL 223, 226.

See eg BVerfG, 1 BvR 971/21, Order of the First Senate (19 November 2021) para 114. See also H Jarass, 'Vorb. vor Art. 1' in H Jarass and B Pieroth (eds), Grundgesetz für die Bundesrepublik Deutschland: Kommentar (16th edn, CH Beck 2020) para 46.

¹³⁵ Jarass, 'Art. 20' (n 134) para 118.

Appellate Body Report, *China—Rare Earths* (n 121) para 5.116.

Appellate Body Report, *EC—Seal Products* (n 103) para 5.169 (reference omitted). See above Pt III.C.

¹³⁸ Meyer (n 91) 67; JA VanDuzer, P Simons and G Mayeda, *Integrating Sustainable* Development into International Investment Agreements: A Guide for Developing Country Negotiators (Commonwealth Secretariat 2013) 246.

Appellate Body Report, Colombia—Textiles (n 103) para 5.72; Appellate Body Report, Argentina—Measures Relating to Trade in Goods and Services, WT/DS453/AB/R (9 May 2016) 140 See above Pt III.B. para 6.234.

necessity test, they could have simply said so: 'actions by a Party that are necessary to protect ...'. Instead, they adopted the 'designed and applied' language. What the Appellate Body stated with respect to Article XX of the GATT also holds true here: 'In the light of the different connecting words used, we consider that a mixing of the different tests under Article XX(b) and Article XX(g), absent of context, would result in an approach that ignores the important distinctions between the various subparagraphs of Article XX'.¹⁴¹

(3) Conclusion

In conclusion, the measure at issue is *designed* to protect public welfare objectives if it is 'not incapable' of protecting them. This presupposes a genuine means—end nexus, akin to the 'relating to' standard known from the WTO exceptions. In the author's view, it involves an assessment of whether the measure is 'capable of contributing' to the stated legitimate public welfare objective, judged at the time when the measure was adopted.

Because the relevant moment in time to determine suitability is when the measure at issue was designed, that is, when the measure was adopted, it is immaterial for the purposes of the design element whether it later turns out to be insufficient to protect the public welfare objective. This is again different to the necessity test, where it is not the contribution as such but the potential of the measure to make a contribution that needs to be present when it is subject to review. 142 Conversely, if the measure does make a contribution to the protection of the public welfare objective, this evidences its suitability, 143 bearing in mind that for some measures, such as measures designed to combat climate change, it may be some time before any such contribution becomes apparent. 144 It follows that host States have a margin of appreciation when it comes to making predictions about the future. As will be seen, States can, therefore, increase the level of protection given to public welfare objectives without the worry that this might be considered to amount to an indirect expropriation.

b) The level of protection

The exemptions are underpinned by the investors' expectations: investors know that they will be subject to, and must comply with, regulatory measures that may interfere with their investments. '[T]he nature and extent of

¹⁴¹ Appellate Body Report, *China—Rare Earths* (n 121) para 5.116.

¹⁴² Appellate Body Report, *Brazil—Retreaded Tyres* (n 125) para 151; Appellate Body Report, *EC—Seal Products* (n 103) paras 5.169, 5.214; Appellate Body Report, *Colombia—Textiles* (n 103) para 5.70. But see Pt III.C.2 re the application of the measure at issue.

For WTO law, see Appellate Body Report, China—Rare Earths (n 121) para 5.100.

¹⁴⁴ cf Appellate Body Report, *Brazil—Retreaded Tyres* (n 125) para 151; Appellate Body Report, *US—Gasoline* (n 83) 21.

governmental regulation or the potential for government regulation in the relevant sector' delineate the investors' horizon of expectations when making the investment. ¹⁴⁵ This, of course, is equally relevant to an FET claim as it is to an expropriation claim. As Khachvani stresses,

each society strikes a certain balance between specific sorts of economic interests, on the one hand, and common interests such as environment and health on the other. When an investor acquires its investment and related property rights in the host State, it accepts that balance as found.¹⁴⁶

This is common ground among investment lawyers. It seems generally accepted that States will not have to compensate in respect of measures which seek to maintain the level of protection for public welfare objectives which existed at the time of the investment.¹⁴⁷ But what if a host State wished to increase the level of protection offered, by, for example, imposing more stringent environmental standards or by re-zoning areas where mining may occur?

An increase of the level of protection raises difficult questions. Was the foreign investor supposed to 'expect' this change in attitude and approach to risk on the part of the host State? Put differently, should the costs of an increase of protection be borne by society as a whole or the individual (here the foreign investor)? Whether measures intended to provide such enhanced protection amount an expropriation depends on whether that increase in protection is covered by the exemptions.

Khachvani denies that they are: 'If the State ... alters the balance [between private interest and public welfare] in favor of public welfare and thereby renders the investor's assets economically unviable, it will have to pay compensation.' Framing the question as a risk allocation issue, Khachvani contends that

The risk that changed circumstances may render a particular activity a nuisance is a risk that an entrepreneur should assume as a business risk. In contrast, the risk that the State may enhance standards of public welfare at the expense of limiting or banning an investor's activity is a sovereign risk. ¹⁴⁹

The counterarguments are the phenomenon of regulatory chill and democratic considerations. A new government that is more environmentally sensitive than a previous government would be considerably hampered by this reading. For example, the environmental concerns of a society may and do change over time. Would this not be covered by the investors' horizon of expectations as well?

In the author's view, the answer turns on whether or not host States are entitled under IIAs to set the levels of protection as under international trade law. As a preliminary point, it should be noted that regulatory chill is largely the result of broad interpretation of the FET standard. A finding of indirect

 $^{^{145}}$ cf fn 36 to para 3(a)(ii) Annex 9-B to the TPP; fn 19 to para 3(a)(ii) Annex 14-B to the USMCA. 146 Khachvani (n 36) 165. 147 ibid 165–6, 171. 148 ibid 165. 149 ibid 166.

expropriation, by contrast, presupposes a substantial deprivation of property interests. 150 Otherwise, the measures in question would not have 'an effect equivalent to direct expropriation', 151 or, in the words of the tribunal in EMV v Czech Republic, 'the measures ... must, however indirect, amount to a "taking". 152 Every encroachment upon property which is below that threshold, a loss in profitability or value for example, would not qualify. 153 This raises the question whether only the total loss of value or control would be deemed 'substantial'. The tribunal in Busta v Czech Republic found that 'loss amounting to two-thirds of the total value ... would represent a substantial loss of economic value', 154 whereas the tribunals in Total v Argentina and Mobil v Venezuela required 'a total loss'. 155 Whilst losses below that high threshold may still fall foul of the FET standard, that is a different issue. 156 The bottom line is that if a regulatory measure does not substantially interfere with property interests, it is simply not expropriatory for that reason alone, and so there is no need to rely upon an exemption. It is important to bear this in mind when considering regulatory chill caused by indirect expropriation.

Under international trade rules, it is settled that not only can States determine the public welfare goals that they wish to pursue (eg environmental protection) but they can also set the level of protection (eg high level of environmental protection exceeding international standards).¹⁵⁷ A complainant could not

151 Para 1(b) Annex 8-A to CETA; para 3 Annex 9-B to the TPP; para 3 Annex 14-B to the USMCA; para 2(b) Annex 10B to RCEP. For the equivalence requirement in general, see Kammerhofer (n 9) 221-8.
 152 European Media Ventures v Czech Republic, UNCITRAL, Partial Award on Liability (8 July

European Media Ventures v Czech Republic, UNCITRAL, Partial Award on Liability (8 July 2009) para 47.

153' LG&E v Argentina (n 98) para 191; Archer Daniels Midland and Tate & Lyle Ingredients Americas v United Mexican States, ICSID Case No ARB(AF)/04/5, Award (21 November 2007) para 248; El Paso Energy International Company v Argentine Republic, ICSID Case No ARB/03/15, Award (31 October 2011) paras 249–56; Crompton (Chemtura) Corporation v Government of Canada, UNCITRAL, Award (2 August 2010) paras 263–5; Charanne and Construction Investments v Spain, SCC Case No 062/2012, Final Award (21 January 2016) para 465; Philip Morris v Uruguay (n 33) para 286. See also Kammerhofer (n 9) 279–80, 284–5; Faccio (n 2) 105, 110–11, 161–2.

154 Busta v Czech Republic, SCC Case No 2015/014, Final Award (10 March 2017) para 390.
155 Total v Argentine Republic, ICSID Case No ARB/04/1, Decision on Liability (27 December 2010) para 195; Mobil Cerro Negro Holding, Mobil Cerro Negro, Mobil Corporation v Bolivarian Republic of Venezuela, ICSID Case No ARB/07/27, Award (9 October 2014) para 286.

156 El Paso v Argentina (n 153) para 459; PSEG Global and Konya Ilgin Elektrik Üretim ve Ticaret Limited Sirketi v Republic of Turkey, ICSID Case No ARB/02/5, Award (19 January 2007) para 238; Enron Creditors Recovery Corporation and Ponderosa Assets v Argentine Republic, ICSID Case No ARB/01/3, Award (22 May 2007) paras 249, 363. See also Faccio (n 2) 162–3.

¹⁵⁷ Appellate Body Report, *EC—Seal Products* (n 103) para 5.200; Appellate Body Report, *Brazil—Retreaded Tyres* (n 125) paras 140, 210; Appellate Body Report, *United States—*

¹⁵⁰ See eg para 1(b) Annex 8-A to CETA; *Pope & Talbot v Canada* (n 65) para 102; *Fireman's Fund Insurance Company v United Mexican States*, ICSID Case No ARB(AF)/02/1, Award (17 July 2006) para 176(c); *Electrabel v Hungary*, Decision on Jurisdiction, Applicable Law and Liability (n 112) para 6.62; *Burlington v Ecuador* (n 31) para 456; *Philip Morris v Uruguay* (n 33) para 192. For the substantial deprivation test in general, see Kammerhofer (n 9) 268–94.

claim that the level of protection afforded is excessive. Provided the public welfare objective is legitimate, its protection cannot be diminished through trade litigation. Government measures can only be challenged on grounds that there is a less trade-restrictive measure that achieves the same public welfare objective to the extent desired by the respondent. Stopping there (ie with necessity), as was the consistent case law of the WTO Appellate Body, removes the balancing component from the proportionality test, because no competing rights and interests are weighed against each other. In sum, there is no strict proportionality test in international trade law.¹⁵⁸ This is true of WTO and free trade agreements (FTAs) alike.¹⁵⁹

It has already been seen that strict proportionality is ruled out under the exemptions, except for 'rare circumstances'. So, under normal circumstances, no compensation would need to be paid if the host State introduced measures to increase the level of protection given to a public welfare objective. Moreover, Khachvani's view leads to a conflict with European Union law. The European Court of Justice held in the *CETA Opinion* that the deep scrutiny of domestic law that ensues from a strict proportionality test (namely, 'call[ing] into question the level of protection of a public interest') is not compatible with EU primary law. ¹⁶⁰

2. Applied to protect

While the design element is concerned with the substance of the measure at issue, the second element relates to the application of the measure, which must be undertaken in the same spirit of protecting public welfare objectives as the underlying measure, and includes implementing administrative regulations and procedures. Considering the substance of a measure and its implementation by domestic authorities separately is likely to have been influenced by WTO law, where the same distinction is made in the chapeau to the general exceptions¹⁶¹ and Article X:3(a) of the GATT. With respect to the latter, the Appellate Body found that 'a distinction must be made between the legal instrument being administered and the legal instrument that

Measures Affecting the Cross-border Supply of Gambling and Betting Services, WT/DS285/AB/R (20 April 2005) para 308; Appellate Body Report, European Communities—Measures Affecting Asbestos and Asbestos-containing Products, WT/DS135/AB/R (5 April 2001) para 168. For the Agreement on Technical Barriers to Trade, see Rec 6 of the Preamble ('at the levels it considers appropriate'). For the WTO Agreement on the Application of Sanitary and Phytosanitary Measures, see fn 3 to art 5(6) ('another measure ... that achieves the appropriate level of sanitary or phytosanitary protection'). For the concept of 'level of protection' in general, see Paine (n 92) 702–3.

¹⁵⁹ For FTAs, see eg arts 23.2, 24.3 CETA; art 17.1 Free Trade Agreement Between Central America, the Dominican Republic and the United States of America (CAFTA-DR).

Opinion 1/17 (n 111) paras 148, 160.

¹⁶¹ Art XX GATT; art XIV General Agreement on Trade in Services. See C Riffel, 'The Chapeau: Stringent Threshold or Good Faith Requirement' (2018) 45 LIEI 141, 162–3.

regulates the application or implementation of that instrument'. ¹⁶² From this it can be inferred that the second part of the test concerns 'the substantive content of legal instruments that regulate the application or implementation of laws, regulations, decisions, and administrative rulings'. ¹⁶³

As with the definition of the design element, the measure must be applied in a manner that does not contradict the public welfare objective. ¹⁶⁴ This implies the *continuous* furtherance of the stated public welfare objective following the adoption of the measure, though bearing in mind that the contribution that the measure makes to that protection need not be evident at the time of review. ¹⁶⁵ The requirement of continuousness is made clearer in Canada's Model FIPA ('maintained'). The good faith test in the Model FIPA does not directly allude to proportionality and might, therefore, entail a lesser degree of scrutiny. However, this alone is unlikely to lead to different results, given the overlap between the principles of good faith and proportionality. ¹⁶⁶ The more significant difference with the 'designed and applied' test arises from the omission in the Model FIPA of the 'rare circumstances' caveat, which will now be considered.

IV. RARE CIRCUMSTANCES

The exemptions limit the concept of indirect expropriation. The arbitral tribunal in *Pope & Talbot* warned of the risk of making that concept inutile through an overly wide application of the police powers doctrine. ¹⁶⁷ For that reason, some IIAs limit the limitations themselves: in 'rare circumstances', the exemptions do not apply. This technique reflects the difficulty of finding the proper calibration of investment protection. ¹⁶⁸

The four mega-regional agreements under consideration come to different solutions: In CETA, the *Schranken-Schranke*—the limitation on the limitation—is confined to one 'rare circumstance', namely 'when the impact of a measure or series of measures is so severe in light of its purpose that it appears manifestly excessive'. This means that a non-discriminatory regulatory measure would be expropriatory (and hence trigger the host State's obligation to compensate) should this exceptional circumstance be met, along with the other requirements of expropriation (eg interference 'with a tangible or intangible property right or property interest'). The CPTPP and

Opinion of Horacio Grigera Naón (9 September 2021) paras 5, 26.

167 Pope & Talbot v Canada (n 65) para 99. See also Malakotipour (n 17) 241.

Appellate Body Report, European Communities—Selected Customs Matters, WT/DS315/AB/R (11 December 2006) para 200.

¹⁶⁴ For WTO law, see Appellate Body Report, EC—Seal Products (n 103) para 5.306; Appellate Body Report, Brazil—Retreaded Tyres (n 125) para 227. 165 See above Pt III.C.1.a(3). 166 cf Eco Oro Minerals v Republic of Colombia, ICSID Case No ARB/16/41, Partial Dissenting

For the regulatory chill resultant from indirect expropriation claims, see Malakotipour (n 17) 245–7, 269–70.

Para 3 Annex 8-A to CETA.

¹⁷⁰ Para 1 Annex 9-B to the TPP; para 1 Annex 14-B to the USMCA; para 1 Annex 10B to RCEP. See Faccio (n 2) 113.

the USMCA refer to 'rare circumstances', in the plural, without specifying what such circumstances might be. 171 One of those circumstances has been identified by the Methanex tribunal: 'unless specific commitments had been given by the regulating government to the then putative foreign investor contemplating investment that the government would refrain from such regulation'. 172 RCEP, like the ASEAN Comprehensive Investment Agreement¹⁷³ and the COMESA Investment Agreement, ¹⁷⁴ omits the 'rare circumstances' proviso altogether and so has the most restrictive concept of indirect expropriation. 175 Under the CPTPP, USMCA, and CETA, the exemptions establish a rebuttable presumption. 176

Looking at the CETA exemption in more detail, that exemption does not apply if, due to its severe impact, the measure is 'manifestly excessive' in light of its purpose. This imports a strict proportionality test. 177 As has been seen, under such a test States cannot set the level of protection they desire; an increase of protection without the payment of compensation would not be sanctioned. However, this presupposes that there has been a substantial deprivation of 'the fundamental attributes of property ..., including the right to use, enjoy and dispose of its investment', having 'an effect equivalent to direct expropriation'. 178 That is, for the rare circumstance exception to be relevant, either the substantial deprivation must be excessive 'in light of its purpose' or the impact of the measure must exceed even a 'substantial' deprivation (ie amount to a total loss of value/control, or close to it). Otherwise, Khachvani would be right in pointing out that 'If the regulation has no such substantial effect, it would not in any event qualify as expropriation and the State would not be under the primary obligation to compensate.'179

Other BITs give similar examples of rare circumstances. For instance, the US-Korea FTA as well as the Trilateral Investment Treaty Between Japan, Korea and China mention the circumstance 'when an action ... is extremely severe or disproportionate in light of its purpose or effect'. 180 The Canada— Colombia FTA limits the exemption 'when a measure ... is so severe in the light of its purpose that it cannot be reasonably viewed as having been

¹⁷¹ Para 3(b) Annex 9-B to the TPP; para 3(b) Annex 14-B to the USMCA.

Methanex v USA (n 63) Pt IV, Ch D, para 7 (emphasis added).

¹⁷³ Para 4 Annex 2 to the ASEAN CIA.

Art 20(8) Investment Agreement for the COMESA Common Investment Area (signed 23 lav 2007, not in force). May 2007, not in force).

176 Malakotipour (n 17) 248; Ortino (n 92) 362.

¹⁷⁷ Prezas (n 54) 159; Ortino (n 92) 363.

Para 1(b) Annex 8-A to CETA. See above Pt III.C.1.b. Lim, Ho and Paparinskis (n 88) 334; Prezas (n 54) 152, 155, 163 refers to the finding of substantial deprivation as the prima facie 179 Khachvani (n 36) 158. definition of indirect expropriation.

Para (3)(b) Annex 11-B to US-Korea FTA; para 2(c) Protocol to the Agreement Among the Government of Japan, the Government of the Republic of Korea and the Government of the People's Republic of China for the Promotion, Facilitation and Protection of Investment (signed 13 May 2012, entered into force 17 May 2014).

adopted in good faith'.¹⁸¹ Finally, the India–Korea Comprehensive Economic Partnership Agreement lists as a rare circumstance 'when a measure ... is extremely severe or disproportionate in light of its purpose or effect'.¹⁸² What those caveats have in common is the requirement of 'a very significant aggravating element or factor in the conduct of the State' that prompts the application of a strict proportionality test.¹⁸³

In the next section, the burden of proof under the exemptions will be discussed.

V. THE BURDEN OF PROOF

As to burden of proof, the UNCITRAL Arbitration Rules 2013 provide that 'Each party shall have the burden of proving the facts relied on to support its claim or defence'. ¹⁸⁴ The same is true for the recently amended ICSID Arbitration Rules. ¹⁸⁵ The CPTPP and the USMCA stipulate that 'the investor has the burden of proving all elements of its claims, consistent with general principles of international law applicable to international arbitration'. ¹⁸⁶ Because the allocation of the burden of proof in the CPTPP and the USMCA reflects a general principle (*actori incumbit onus probatio*), ¹⁸⁷ the same applies to the other mega-regionals under investigation which do not have such an explicit rule.

It follows that the claimant investor must make a prima facie case that the measure at issue is expropriatory. This involves making a case that there has been a substantial deprivation of property interests. What is not clear is whether the investor's burden of proof also extends to the exemptions. The affirmative view is predicated upon the belief that exemptions constitute negative elements of the claim. As a consequence, the claimant investor would have to show that the measure at issue was *not* designed and applied to protect the legitimate public welfare objective pursued by the respondent. See law in relation to the police powers doctrine seems to confirm this. In *Servier v Poland*, the tribunal found that 'the burden ... falls onto the Claimants to show that Poland's regulatory actions were *inconsistent* with a legitimate exercise of Poland's police powers. If the Claimants produce

¹⁸¹ Para 2(b) Annex 811 to Canada-Colombia Free Trade Agreement (signed 21 November 2008, entered into force 15 August 2011).

Para 3(b) Annex 10-A to India-Korea CEPA.

¹⁸³ Eco Oro v Colombia (n 32) paras 643, 698. 184 Art 27(1). 185 Rule 36(2).

Art 9.23.7 TPP; para 1 Annex 14-E to the USMCA in conjunction with art 14.D.7.7.

¹⁸⁷ See also Appellate Body Report, *United States—Measure Affecting Imports of Woven Wool Shirts and Blouses from India*, WT/DS33/AB/R and Corr.1 (23 May 1997) 14 ('the burden of proof rests upon the party, whether complaining or defending, who asserts the affirmative of a particular claim or defence'); *Metal-Tech v Republic of Uzbekistan*, ICSID Case No ARB/10/3, Award (4 October 2013) para 237; *Tokios Tokelés v Ukraine*, ICSID Case No ARB/02/18, Award (26 July 2007) para 121. For burden of proof in investment arbitration in general, see K Reichert et al, 'Proof: Investment Arbitration' in H Ruiz Fabri (ed), *Max Planck Encyclopedia of International Procedural Law* (OUP 2020) paras 10–28.

188 Malakotipour (n 17) 266.
189 For the distribution of the burden of proof under exceptions in IIAs, see Baetens (n 70) 164.

sufficient evidence for such a showing, the burden shifts to Poland to rebut it.' ¹⁹⁰ In a similar vein, the WTO Panel in *EC—Approval and Marketing of Biotech Products* allocated the burden of showing that the requirements of Article 5.7 of the Agreement on the Application of Sanitary and Phytosanitary Measures, which it qualified as an exemption, were *not* met to the complainant. The Panel held that

Characterizing Article 5.7 as a qualified right and not an exception also has implications for the allocation of the burden of proof concerning the issue of the consistency of an SPS measure with Article 5.7 ... [I]n cases where a complaining party alleges that an SPS measure is inconsistent with the obligation in Article 2.2 not to maintain SPS measures without sufficient scientific evidence, it is incumbent on the complaining party, and not the responding party, to demonstrate that the challenged SPS measure is inconsistent with at least one of the four requirements set forth in Article 5.7. If such non-compliance is demonstrated, then, and only then, does the relevant obligation in Article 2.2 apply to the challenged SPS measure. ¹⁹¹

The opposing view takes the 'rare circumstances' caveat as a starting point. As has been seen, when the requirements of the exemptions are satisfied, they establish a presumption against the regulatory measure being expropriatory. The claimant investor can rebut that presumption by invoking a rare circumstance (in which case the measure would be deemed expropriatory). This means that the investor bears the burden of proving that the situation amounts to a 'rare circumstance'. 193 From this, it can be deduced that the burden of proof with respect to the exemptions rests with the respondent State. It is for the respondent to prove their assertion that the measure at issue is a non-discriminatory regulatory action rather than an act of expropriation. Otherwise, the investor would bear the burden of proof for both the rule (ie the presumption) and the exception (ie its rebuttal).

Reflecting general principles of international law, the burden of proof regarding exemptions should be allocated as follows: the State should show that the measure at issue is of general application and pursues a legitimate public welfare objective. Should this be achieved, the burden then shifts to the investor to show that there is a 'rare circumstance' for their claim to be able to proceed.

VI. THE OBLIGATION TO COMPENSATE

It is undisputed that a foreign investor is entitled to compensation even for lawful expropriation, be it direct or indirect. 194 As a corollary, compensation

Les Laboratoires Servier, Biofarma, Arts et Techniques du Progres v Republic of Poland, UNCITRAL, Award (14 February 2012) para 584 (emphasis added).

Panel Report, EC—Approval and Marketing of Biotech Products (n 41) para 7.2976.
 See above Pt IV.
 Prezas (n 54) 164.

¹⁹⁴ Art 8.12.1(d) CETA; art 9.8.1(c) TPP; art 14.8.1(c) USMCA. See also Compañia del Desarrollo de Santa Elena v Costa Rica (n 60) para 68; Franz Sedelmayer v Russian Federation, SCC, Arbitration Award (7 July 1998) 73.

is also due when an (eg discriminatory) expropriation is lawful as a result of an exception. At first blush, this conclusion seems counter-intuitive. It is relevant in relation to RCEP, where the general exceptions capture expropriation. 195 In the author's view, what the Santa Elena tribunal stated with respect to the environment can be applied to other legitimate public welfare objectives, that is, potential grounds of justification: 'Expropriatory environmental measures —no matter how laudable and beneficial to society as a whole—are, in this respect, similar to any other expropriatory measures that a state may take in order to implement its policies.'196 In such a situation, 'society as a whole [and not the expropriated owner] should bear the economic burden of achieving' a public welfare objective; and further, 'society cannot advance its welfare at the expense of a few individuals when the cost for such advancement should be borne by society as a whole'. 197

The difference between this and unlawful expropriation is the kind (and amount) of compensation owed: fair market value if the expropriation was lawful, ¹⁹⁸ if not, the investor's expectation interest (ie 'any financially assessable damage including loss of profits insofar as it is established'). 199 The tribunal in Bear Creek Mining v Peru confirmed, in connection with an 'unlawful indirect expropriation', that a general 'exception ... does not offer any waiver from the obligation ... to compensate for the expropriation', and that 'damages for an unlawful expropriation should at least be as much as the compensation for a lawful expropriation'. 200 Conversely, if there is no expropriation—because the requirements of an exemption are met—no compensation needs to be paid under that head of the claim. The tribunal in Eco Oro v Colombia considered this outcome an 'inconsistency'. 201

In a nutshell, the issue is whether there is an obligation to provide reparation even if wrongfulness is precluded. Article 27 of the ILC's Articles on State Responsibility are silent on 'the question of compensation for any material loss caused by the act in question', 202 and are in any case confined to circumstances precluding wrongfulness as set out in those Articles.²⁰³ In *Eco* Oro, decided under the Canada-Colombia FTA, the arbitral tribunal denied that there was an (indirect) expropriation²⁰⁴ but still found an obligation to compensate for a breach of the minimum standard of treatment even though

¹⁹⁵ Art 17.12 RCEP. The general exceptions do not apply to expropriation in the other megaregionals under investigation, see art 28.3 CETA, art 29.1 TPP, art 32.1 USMCA.

Compañia del Desarrollo de Santa Elena v Costa Rica (n 60) para 72.
 Khachvani (n 36) 164.

¹⁹⁸ Art 9.8.2(b) TPP; art 14.8.2(b) USMCA; art 10.13.2(b) RCEP; art 8.12.2 CETA.

¹⁹⁹ Art 36(2) Articles on State Responsibility.

Bear Creek Mining Corporation v Republic of Peru, ICSID Case No ARB/14/21, Award (30 ovember 2017) paras 449, 477, 596.

Rtt 50(2) Mittoles on State Responsionity.

Eco Oro v Colombia (n 32) para 831. November 2017) paras 449, 477, 596.

Art 27(b) Articles on State Responsibility. Cf LG&E v Argentina (n 98) para 260; J Kurtz, The WTO and International Investment Law: Converging Systems (CUP 2016) 217.

²⁰³ For the difference between exemptions and circumstances precluding wrongfulness, see Eco Oro v Colombia (n 32) para 699. Viñuales (n 40) 78.

that breach was justified under an exception.²⁰⁵ On that basis, measures necessary to protect legitimate public welfare objectives are subject to compensation.²⁰⁶

In order to determine the legal effects of the exception, the tribunal drew upon the wording of the exemption in the annex on expropriation ('non-discriminatory measures ... do not constitute indirect expropriation').²⁰⁷ Contrasting the exception with the exemption, the tribunal concluded that, because the former does not stipulate 'the circumstances in which a measure is not to constitute a treaty breach', its successful invocation does not rule out the claimant's entitlement to compensation.²⁰⁸ This author disagrees. In truth, the differences in wording are intended to distinguish exemptions from exceptions and by no means imply that only the former exclude liability.

A. Difference between Exemption and Exception

The tribunal in *Eco Oro* should have paid heed to the difference between exemption and exception: exemptions mean that there has not been an (indirect) expropriation, exceptions offer a justification for what would otherwise have been an illegal expropriation, thus making it lawful.²⁰⁹ An exception only comes into play if there is an expropriation,²¹⁰ whereas if the conditions for an exemption are met, the measure is not an expropriation and so does not need to be justified.²¹¹

The *Eco Oro* tribunal misunderstood the purpose of general exceptions when it said that 'neither environmental protection nor investment protection is subservient to the other, they must co-exist in a mutually beneficial manner'. This may be correct in relation to obligations in IIAs which have to be interpreted in harmony with other international law obligations of the parties (as per Article 31(3)(c) of the VCLT); it is wrong in relation to exceptions. It amounts to a reversal of the logic of exceptions and defies their *ratio legis*, which is to relieve a respondent of liability. When the conditions of an exception clause are satisfied, the public welfare objective protected under the exception prevails: 'nothing in this Agreement shall be construed to prevent a Party from adopting or enforcing measures ...'. There is no place for co-existence. That 'the FTA is equally supportive of investment protection', as postulated by the tribunal, 215 is irrelevant under an exception. The salient

```
205 ibid, para 830.
207 Para 2(b) Annex 811 to Canada–Colombia FTA.
208 Eco Oro v Colombia (n 32) paras 829–30, 832, 837.
210 See Viñuales (n 40) 66, 68, 73, 79.
211 Eco Oro v Colombia (n 32) para 828.
213 For defences in general, see J Sharpe and M Jacob, 'Counterclaims and State Claims' in C

Beharry (ed), Contemporary and Emerging Issues on the Law of Damages and Valuation in International Investment Arbitration (Brill 2018) vol 11, 348. See also Rajput (n 4) 188.
214 See eg art 2201(3) Canada–Colombia FTA.
215 Eco Oro v Colombia (n 32) para 829.
```

point is that a justified breach is lawful.²¹⁶ In this context, expropriation has to be distinguished from other substantive investment protection standards. As regards a justified and hence lawful expropriation, it is worth recalling that, in the author's view, compensation would still be due, not to the tune of the investor's expectation interest but limited to the fair market value.

B. Permission to Act

The *Eco Oro* tribunal makes a distinction between the right of the respondent to take a measure (under an exception), on the one hand, and the liability that flows from doing so, on the other.²¹⁷ The fundamental fallacy is that the tribunal assumed that permission to take action 'without the payment of compensation' needs to be expressly articulated in IIAs,²¹⁸ whereas it is the opposite that is the 'default position under general international law'.²¹⁹ Under the Articles on State Responsibility, the precondition for compensation is an 'internationally wrongful act',²²⁰ which the successful plea of an exception negates. What is more, the right of States to expropriate has long been recognized.²²¹ Contrary to *Eco Oro*, other tribunals have held that 'where the bilateral investment treaty at hand contains an exculpatory provision ..., compensation is not payable to the extent that such provision exonerates that party from liability',²²² and with regard to a security exception, that 'it is the factor excluding the State from its liability vis-à-vis the damage caused as a result of the measures adopted by Argentina in response to the severe crisis suffered by the country'.²²³

This line of case law accords with WTO law where it is uncontested that a WTO Member may infringe its WTO obligations *without legal consequences* if the requirements of an exception clause are met. It is surprising that the tribunal disregarded standing WTO jurisprudence, given that the pertinent FTA, in a different part, explicitly requires its consideration for the purposes of interpreting similarly worded provisions.²²⁴ In the final analysis, the conclusion must be that there is no obligation on the host State to compensate the foreign investor for measures that fall within the scope of an exemption.

Henckels, 'Scope Limitation or Affirmative Defence?' (n 40) 363. For circumstances precluding wrongfulness, see F Paddeu, 'Clarifying the Concept of Circumstances Precluding Wrongfulness (Justifications) in International Law' in F Paddeu and L Bartels (eds), Exceptions in International Law (OUP 2020) 206, 224.
 Eco Oro v Colombia (n 32) para 832.
 ibid, para 829.

²²⁰ Art 36(1) Articles on State Responsibility.

Amoco v Iran (n 58) paras 115, 147; Dolzer and Schreuer (n 58) 98; Radi (n 57) 152.

²²² BG Group v Republic of Argentina, UNCITRAL, Final Award (24 December 2007) fn 327 to para 409.

²²³ LG&E v Argentina (n 98) paras 259–61, 264. Contra CMS Gas Transmission v Argentine Republic, ICSID Case No ARB/01/8, Award (12 May 2005) para 392 ('any suspension of the right to compensation is strictly temporary, and ... this right is not extinguished by the crisis events').

²²⁴ Art 2102(2), 1st sentence, Canada–Colombia FTA. See also Baetens (n 70) 165 ('Whether an explicit reference to GATT/GATS is included or not, the effect would seem to be the same').

VII. CONCLUSIONS

It is the prerogative of the State to regulate in the public interest through the adoption of non-discriminatory public welfare measures. Case law which promotes a broad application of the concept of indirect expropriation constitutes a threat to that prerogative.²²⁵ As so often, the solution lies in careful treaty-drafting. The parties to modern mega-regional trade and investment agreements have, consequently, sought to reduce the discretionary powers of investment tribunals by better delineating the concept of expropriation.²²⁶ To that end, they have appended annexes on expropriation to their IIAs.

The last paragraphs of those annexes are exemptions embodying the police powers doctrine. They restrict the scope of indirect expropriation. If their requirements are met, there is no expropriatory measure, and hence no obligation to compensate. Exemptions constitute negative elements of an expropriation claim.

The burden of proof to show that its actions fall within the exemptions lies with the respondent State. If the claimant investor then wishes to claim that there is, exceptionally, a 'rare circumstance', then this is for the investor to prove. Exemptions incorporate a suitability test, which also considers whether the measure in question has the potential to contribute towards a legitimate public welfare objective. This test is less stringent than the fully-fledged necessity test as found in WTO law and is, therefore, more deferential to State sovereignty.

The suitability of a measure should be assessed as at the time of its adoption. This gives host States ample discretion to take measures, subject to the proviso that the measures are applied to further a legitimate public welfare objective. Where the measures taken fall within the 'rare circumstances' exception, a strict proportionality test applies, meaning that the level of protection of a public welfare objective set by the respondent State can be challenged by the claimant investor as being excessive. This does not mean that the State would be legally obliged to reduce the level of protection; instead, if successful, the claimant would be entitled to compensation for the loss suffered.²²⁷ However, not all IIAs contain such a caveat, it being dependent upon the contracting parties' preference for how strong a level of investment protection they wish to see. In the absence of the 'rare circumstances' caveat, and leaving aside a potential FET claim, foreign investors are not protected against the adverse economic effects of regulatory measures, understood as public welfare measures of general application, on their investments.²²⁸

²²⁵ See eg *Servier v Poland* (n 190) para 577, ruling that dispossession need not be permanent in order to find expropriation.

²²⁶ For CETA, see para 6(c) Joint Interpretative Instrument.

²²⁷ Note that the respondent may pay monetary damages in lieu of restitution of property under the CPTPP (art 9.29.1(b) TPP), CETA (art 8.39.1(b)), and the USMCA (art 14.D.13.1(b)).

²²⁸ Kammerhofer (n 9) 258, 260.